Author's response to reviews

Title: Brucella spp. infection in large ruminants in an endemic area of Egypt: cross-sectional study investigating seroprevalence, risk factors and livestock owner's knowledge, attitudes and practices (KAPs).

Authors:

Hannah R Holt (hholt@rvc.ac.uk)
Mahmoud M Eltholth (meltholth@rvc.ac.uk)
Yamen M Hegazy (yhegazy@rvc.ac.uk)
Wael F El-Tras (wael_tras@yahoo.com)
Ahmed A Tayel (Tayel_Ahmad@yahoo.com)
Javier Guitian (jguitian@rvc.ac.uk)

Version: 4 Date: 9 May 2011

Author's response to reviews:

The formatting changes requested by the editors regarding formatting of the manuscript and changes to references have been made.